Author's response to reviews

Title: Skin pigmentation, sun exposure and vitamin D levels in children of the Avon Longitudinal Study of Parents and Children

Authors:

Carolina Bonilla (C.Bonilla@bristol.ac.uk)
Andrew R Ness (Andy.Ness@bristol.ac.uk)
Andrew K Wills (A.K.Wills@bristol.ac.uk)
Debbie A Lawlor (D.A.Lawlor@bristol.ac.uk)
Sarah J Lewis (S.J.Lewis@bristol.ac.uk)
George Davey-Smith (KZ.Davey-Smith@bristol.ac.uk)

Version: 2 Date: 27 March 2014

Author's response to reviews:

See attached.